Author's response to reviews

Title: Quality of reporting of trial abstracts needs to be improved: using the CONSORT for abstracts to assess the four leading Chinese medical journals of traditional Chinese medicine

Authors:

Ling Wang (linewang84@163.com)
Yulin Li (lee-yulin@163.com)
Jing Li (lijing68@hotmail.com)
Mingming Zhang (mingming-zhang@163.com)
Lin Xu (xulin@163.com)
Wenming Yuan (yuanwenming2000@163.com)
Gang Wang (wong-gang@126.com)
Sally Hopewell (hopewell@cochrane.co.uk)

Version: 3 Date: 25 June 2010

Author's response to reviews:

1. We have corrected the English language, grammar and spelling within the manuscript.
2. We have added a competing interests section at the end of the manuscript, before the reference list as follows:
   This research is supported by a project of the Cochrane Collaboration titled "Making Cochrane review more accessible to policy makers", but no funding for the publication. Dr Sally Hopewell is a lead author on the CONSORT extension for reporting abstracts of randomized trials.
3. We have added an Authors’ Contributions section at the end of the manuscript, before the reference list as follows:
   LW and YL participated in the design of the study, carried out the hand searching of journals, assessment of the abstracts and drafting the manuscript.
   JL, MZ and SH conceived the study, interpreted the data and revised the manuscript.
   LX and WY participated in the design of the study and performed the data analysis.
   GW participated in the design of the study, selection of journals and interpretation of the data.
   All authors read and approved the final manuscript.